Author's response to reviews

Title: Which is the Best Postoperative Chemotherapy Regimen in Patients with Rectal Cancer after Neoadjuvant Therapy?: Analysis of Surveillance, Epidemiology, and End Results-Medicare Data

Authors:

Peng Gao (gaokaji16@163.com)
Yong-xi Song (songyongxi840309@126.com)
Jing-xu Sun (sun2003999@163.com)
Xiao-wan Chen (chenxiaowan516@gmail.com)
Ying-ying Xu (lovecmustar@163.com)
Jun-hua Zhao (zjh900521@163.com)
Xuan-zhang Huang (qq_347310502@126.com)
Hui-mian Xu (xuhuimian@126.com)
Zhen-ning Wang (josieon826@sina.cn)

Version: 2
Date: 14 May 2014

Author's response to reviews:

Annie Lyn Bravo
Journal Editorial Office
BioMed Central

Dear Annie,

Thank you very much for your comments and suggestions on improving our manuscript “Which is the Best Postoperative Chemotherapy Regimen in Patients with Rectal Cancer after Neoadjuvant Therapy?: Analysis of Surveillance, Epidemiology, and End Results-Medicare Data” (1605246167129326). Those comments are all valuable and very helpful for revising and improving our paper.

Following Editor’s comments and suggestions:

1. Following Editor “Please revise your manuscript to include line and page numbers.” Following your advice, we have added line and page numbers in the revised manuscript.

2. Following Editor “A statement detailing what ethical approval has been obtained, including the name of the ethics committee and the reference number where appropriate, must appear in all manuscripts.” Following your advice, we have added a ETHICS STATEMENT at the end of our manuscript before the REFERENCES part.

Thank you very much!

Sincerely yours,
Zhen-ning Wang, M.D., Ph.D.
Professor and Associate Director
Department of Surgical Oncology and General Surgery#
First Hospital of China Medical University
Shenyang 110001, China